---
title: "Daily Updates"
description: "OpenSRE engineering daily updates from merged pull requests"
---

Daily updates are generated each evening (Europe/London) from the pull requests merged that day.

## Latest Updates

| Date | Highlights |
| ---- | ----------- |
| 2026-05-05 | [View](/daily-updates/2026-05-05) — feat(integrations): support Cursor Agent CLI as... |
| 2026-05-04 | [View](/daily-updates/2026-05-04) — feat(cli): allow switching toolcall model at runtime (#1182) |
| 2026-05-03 | [View](/daily-updates/2026-05-03) — feat(integrations): add VictoriaLogs integration (#126) |
| 2026-05-02 | [View](/daily-updates/2026-05-02) — feat(cli): make interactive shell documentation-aware... |
| 2026-05-01 | [View](/daily-updates/2026-05-01) — chore: refactor integrations |
| 2026-04-30 | [View](/daily-updates/2026-04-30) — Issue/319 splunk integration |

## Archive

| Date | Link |
| ---- | ---- |
| 2026-04-29 | [View](/daily-updates/2026-04-29) |
| 2026-04-28 | [View](/daily-updates/2026-04-28) |
| 2026-04-27 | [View](/daily-updates/2026-04-27) |
| 2026-04-26 | [View](/daily-updates/2026-04-26) |
| 2026-04-25 | [View](/daily-updates/2026-04-25) |
| 2026-04-24 | [View](/daily-updates/2026-04-24) |
| 2026-04-23 | [View](/daily-updates/2026-04-23) |
| 2026-04-22 | [View](/daily-updates/2026-04-22) |
| 2026-04-21 | [View](/daily-updates/2026-04-21) |
| 2026-04-20 | [View](/daily-updates/2026-04-20) |
| 2026-04-19 | [View](/daily-updates/2026-04-19) |
| 2026-04-18 | [View](/daily-updates/2026-04-18) |
| 2026-04-17 | [View](/daily-updates/2026-04-17) |
| 2026-04-16 | [View](/daily-updates/2026-04-16) |
| 2026-04-15 | [View](/daily-updates/2026-04-15) |
| 2026-04-14 | [View](/daily-updates/2026-04-14) |
| 2026-04-13 | [View](/daily-updates/2026-04-13) |
| 2026-04-12 | [View](/daily-updates/2026-04-12) |
| 2026-04-11 | [View](/daily-updates/2026-04-11) |
| 2026-04-10 | [View](/daily-updates/2026-04-10) |
| 2026-04-09 | [View](/daily-updates/2026-04-09) |
| 2026-04-08 | [View](/daily-updates/2026-04-08) |
| 2026-04-07 | [View](/daily-updates/2026-04-07) |
| 2026-04-06 | [View](/daily-updates/2026-04-06) |
| 2026-04-05 | [View](/daily-updates/2026-04-05) |
| 2026-04-04 | [View](/daily-updates/2026-04-04) |
| 2026-04-03 | [View](/daily-updates/2026-04-03) |
| 2026-04-02 | [View](/daily-updates/2026-04-02) |

